**Author’s response to reviews**

**Title:** Effect of metastatic site on survival in patients with neuroendocrine neoplasms (NENs). An analysis of SEER data from 2010 -2014

**Authors:**

Nikolaos Trikalinos (ntrikalinos@wustl.edu)

Benjamin Tan (btan@wustl.edu)

Manik Amin (manikamin@wustl.edu)

Jingxia Liu (esther@wustl.edu)

Ramaswamy Govindan (rgovindan@wustl.edu)

Daniel Morgensztern (danielmorgensztern@wustl.edu)

**Version:** 3 **Date:** 05 Mar 2020

**Author’s response to reviews:**

1. The link to your data provided in the Availability of Data and Materials statement is currently broken. Please amend this in the revised version of the manuscript.

Response: We have corrected the link

2. In the availability of data and materials section, please also clarify how to request access to the data.

Response: We have clarified that

3. Please clarify whether any administrative permissions were required to access the raw data. If so please state who granted permission in the “Ethics approval and consent to participate” section.

Response: We have clarified that in the "Availability of Data and Materials" section. We kept N/A as a response in the “Ethics approval and consent to participate” section.

4. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: Have done so. Have changed the color of the text to a uniform black.